Author's response to reviews

Title: Canadian oncogenic human papillomavirus cervical infection prevalence: Systematic review and meta-analysis

Authors:

Andrea C Tricco (triccoa@smh.ca)
Carmen H Ng (carmen_hm_ng@yahoo.ca)
Vladimir Gilca (vladimir.gilca@ssss.gouv.qc.ca)
Andrea Anonychuk (andrea.m.anonychuk@gsk.com)
Ba' Pham (ba.pham@theta.utoronto.ca)
Shirra Berliner (shirraacremer@hotmail.com)

Version: 5 Date: 1 September 2011

Author's response to reviews:

Dear BMC,

In this version, I changed my work phone number extension, two words in the abstract, and five words in the manuscript. Everything looks great now.

Thank you,

Andrea